



Limnology and Oceanography Letters 2021 © 2021 The Authors. Limnology and Oceanography published by Wiley Periodicals LLC. on behalf of Association for the Sciences of Limnology and Oceanography. doi: 10.1002/lol2.10181

## LETTER

# On thin ice: Linking elevation and long-term losses of lake ice cover

Kyle R. Christianson ,<sup>1</sup>\* Kelly A. Loria,<sup>2</sup> Peter D. Blanken,<sup>3</sup> Nel Caine,<sup>1</sup> Pieter T. J. Johnson <sup>2</sup>

<sup>1</sup>Institute of Arctic and Alpine Research, University of Colorado, Boulder, Colorado; <sup>2</sup>Department of Ecology and Evolutionary Biology, University of Colorado, Boulder, Colorado; <sup>3</sup>Department of Geography, University of Colorado, Boulder, Colorado

#### Scientific Significance Statement

Climate change is expected to have particularly strong effects in colder regions, such as those at high elevations. Yet due to remote locations and sparse data, it is unknown how mountain lakes are actually changing compared to other lakes in other regions. By assessing ice phenology trends for one of the highest (> 3000 m) and longest-running (36 yr) datasets on mountain lakes, we find that alpine lakes are decreasing ice cover 50% faster than average for more well-studied lakes in the Northern Hemisphere, emphasizing the importance of studying mountain lakes that provide water to nearly half of the world's population.

## Abstract

Despite long-term analyses of lake ice phenology globally, comparatively little is known about high-elevation lakes, for which climate shifts are thought to be occurring faster than at lower elevations. Using a 36-yr dataset (1983–2018) on alpine lakes (> 3000 m ASL) from the Green Lakes Valley, Colorado (GLV), we found that ice-cover duration decreased by an average of ~ 24 d, due to both earlier ice-off (9 d) and especially later ice-on (15 d). Spring ice thickness also decreased by 0.88 cm yr<sup>-1</sup>. By comparison, ice-cover duration in the GLV is decreasing ~ 50% faster than for Northern Hemisphere lakes (*n* = 215), which translates to an increase in open water duration ~ 2.5 times more in the GLV than the average of the Northern Hemisphere. Our analytical comparison demonstrated more rapid trends in this alpine region compared to lakes more broadly, and especially emphasized the influence of elevation on lake ice phenology.

\*Correspondence: christianson.kyle@colorado.edu

Associate editor: Peter Leavitt

**Authorship Contribution Statement:** All authors contributed to the manuscript and study design. KRC, KAL, and PTJJ performed analyses. TNC and KAL collected and organized data. KRC wrote the manuscript, while KAL, PDB, TNC, and PTJJ provided edits, comments, and contributions to the final manuscript.

**Data Availability Statement:** Data are available in the Niwot Ridge LTER repository at https://niwot-ridge-lter.squarespace.com/data-catalog

Additional Supporting Information may be found in the online version of this article.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited. As integrators of atmospheric processes, lakes are influenced by shifts in climate (Brown and Duguay 2010). For example, increased air temperatures influence lake temperatures and thermal stability (Christianson et al. 2020), which can alter biotic interactions (Jacobson et al. 2008) and community composition (Arvola et al. 2010). In temperate lakes, altered ice phenology is an indicator of changing lake conditions (Sharma et al. 2019). Decreased ice-cover duration can shift or extend lake mixing regimes (Preston et al. 2016), alter zooplankton abundance (Adrian et al. 1999), affect ecosystem processes (Caldwell et al. 2020), increase metabolic demands of ectotherms (Christianson and Johnson 2020), and alter freshwater resources (Sharma et al. 2019).

Ongoing climate shifts are associated with changes in lake ice phenology worldwide (Sharma et al. 2019). Changes in air temperature and precipitation can alter these patterns and lead to long-term changes in ice phenology (Brown and Duguay 2010). Other factors, such as lake geography (Williams and Stefan 2006), can have important mediating effects for local and regional differences in ice-cover duration. For example, Williams and Stefan (2006) showed that higher latitudes and elevations can delay lake ice clearance and advance ice formation, such that lake ice-cover duration increases by ~ 3.7 d per degree north and ~ 0.04 d per meter in elevation. Although climatic change effects are greater for lakes in cold regions (Williamson et al. 2009), there are fewer long-term empirical data on patterns in lake ice phenology comparing these regions to global trends. Multiple lines of evidence suggest that ice phenology is likely to shift more rapidly and to a greater relative extent in colder regions. Processes such as Arctic amplification and snow/ice-albedo feedback (Henneman and Stefan 1999) can lead to faster changes in lakes in cold regions relative to other regions.

Due to their typically remote locations, mountain lakes are also a generally understudied class of lakes (Catalan and Donato-Rondón 2016), and the paucity of long-term empirical data on alpine lake ice-cover duration has limited efforts to compare trends in ice phenology among lakes globally. The range in values for alpine/montane systems are often defined regionally and less applicable broadly. For example, an elevation of 265 m ASL (above sea level; Sharma et al. 2019) or 608 m ASL (Kainz et al. 2017) would not be representative of high-elevation regions in the western United States, where lakes above 2000 m ASL are common (Bahls 1992). Thus, ice phenology trends for many high-elevation lakes remain unknown and are likely underestimated. Ice phenology trends are also sensitive to the timing and length of the sampling period, such that analyses comparing patterns of ice phenology across elevation gradients should be compared using the same time period (Kainz et al. 2017).

To better understand the patterns of ice-phenology change in alpine lake ecosystems and how they differ from lakes across elevational and latitudinal gradients generally, we analyzed temporal trends in ice-on date, ice-off date, ice thickness, total ice-cover duration, and heterogeneity for lakes in the Green Lakes Valley (GLV) of the Southern Rocky Mountains, which represent the longest-running high-elevation study on lake ice globally (51 yr for seven lakes at 3126-3620 m ASL). This study extends the dataset for lakes in the GLV by 3 yr for ice-off, 12 yr for ice-on, and 16 yr for ice thickness relative to any previously published research (Caine 2002; Williams and Stefan 2006; Preston et al. 2016). In addition, we compared results from the GLV with those from natural lakes in the Northern Hemisphere along gradients in both latitude and elevation, with the aim of using a consistent analytical framework to assess the magnitude and rate of ice cover changes for the period from 1983 to 2013. Our goals were to determine the degree to which phenological shifts in ice cover associated quantitatively with lake elevation and latitude, thereby comparing trends in cold regions to those more broadly.

## **Methods**

## **Green Lakes Valley**

Research in the GLV represents the longest-running study (51 yr) of high-elevation lakes (> 3000 m ASL) in North America. Located in north-central Colorado within the Southern Rocky Mountains and just east of the Continental Divide (Fig. 1), the GLV includes a series of seven montane to alpine lakes between 3126 and 3620 m ASL, with maximum depths ranging 7-16 m, surface areas 0.4-2.2 km<sup>2</sup>, and mean depths 3.7-8.6 m (Caine 2001). The historical mean annual air temperature for the GLV is about  $-4.1^{\circ}$ C (range: -7.7 to  $-0.5^{\circ}$ C), while annual precipitation averages 993 mm (range: 512-1581 mm) (Caine 2001). Since 1968, ice clearance date has been recorded in Green Lake 4 (GL4), while ice clearance and formation date have been recorded throughout the GLV since 1982. Based on weekly to biweekly observations throughout spring and fall, ice formation and clearance dates were classified visually as when each lake was 100% covered or cleared of ice, respectively. Ice thickness has also been measured since 1982 in GL4 (Caine 2018) by inserting a 320-cmlong probe into a drilled ice hole at the deepest part of the lake between 20 and 31 March, when maximum ice thickness typically occurs. These measurements were collected through "black" ice and did not include snow (see Caine 2002). Ice thickness data are available in the Niwot Ridge LTER repository at https://portal.edirepository.org/nis/ mapbrowse?packageid=knb-lter-nwt.199.2. For this study, we included Green Lakes 1-5 (3400-3620 m ASL) and years in each lake for which ice clearance and the previous year's ice formation were recorded. Ice phenology data are available in the Niwot Ridge LTER repository at https://portal. edirepository.org/nis/mapbrowse?packageid=knb-lter-nwt.106.3 (Caine 2018). This generated a dataset from 1983 to 2018. Due to the importance of open water duration for temperate lakes, we (1) analyzed trends in the duration of open water (ice-free days) using the complete GLV dataset and (2) trends in icecover duration (ice-covered days) for comparisons with an available database on Northern Hemisphere lakes (1983-2013) (Benson et al. 2000).

## Northern Hemisphere comparisons

To analyze patterns in lake ice cover in the Northern Hemisphere as a function of latitude and elevation, we obtained data from the Global Lake and River Ice Phenology Database housed at the National Snow and Ice Data Center. Data and metadata are available at http://nsidc.org/data/g01377.html (Benson et al. 2000). These data encompass centuries of ice records and represent the most comprehensive global dataset in lake ice cover. We limited our analysis to Northern Hemisphere's natural lakes that freeze annually, excluding rivers or streams. We further constrained the analysis to the period between 1983 and 2013, which maximized overlap with the GLV dataset while including the most recent update to the global database at time of submission. Further analysis of



Fig 1. Locations of lakes in the Northern Hemisphere dataset (top; red) from 1983 to 2013 and the Green Lakes Valley, Colorado (bottom). Green Lakes 1–5 are designated as GL1–5, with Lake Albion also shown. The continental divide is designated with a dashed line.

the entire Northern Hemisphere dataset can be found in Sharma et al. (2019) and Lopez et al. (2019). For each record, we gathered information on the elevation, latitude, year, and ice-cover duration of each lake, predefined in the dataset as the time between ice formation and ice clearance. In total, we obtained 3221 observations representing 215 lakes and an elevational range of 5–1768 m ASL, latitudinal range of  $36.2-74.4^{\circ}N$  (Fig. 1), and for lakes with reported area and depth, areas ranged  $0.005-31,500 \text{ km}^2$  (mean:  $445 \text{ km}^2$ , median:  $15 \text{ km}^2$ ), and maximum depths 1.8-1741 m (mean: 50 m, median: 27 m).

#### Statistical analyses

We used linear mixed-effects models (LME) using the lme4 package (Bates et al. 2015) in R 3.6.0 (R Core Team 2019) to determine trends in ice clearance, ice formation, and open water duration in the GLV. The number of days until ice-off, ice-on, or total days of ice coverage or open water was modeled as a Gaussian distributed response variable, with elevation and sampling year as fixed effects. To account for repeated (nonindependent) observations on each lake, we incorporated lake identity as a random intercept term. We also accounted for random slopes by allowing the annual trend in ice duration to vary by lake. Similarly, the Northern Hemisphere dataset was analyzed to determine ice-cover duration trends while accounting for fixed effects of elevation, latitude, and year, again incorporating a random intercept term for lake identity and random slopes for year. Although other measures such as lake size, depth, and volume may be important, these values are less commonly available and were not found to be significant during initial analysis. Because ice thickness was observed in only a single lake (i.e., GL4), we first used a linear regression model to determine the yearly trend; after further consideration, we applied a break-point analysis to account for potential nonlinearity using the "segmented" package (Muggeo 2020) in R. Heterogeneity of ice-cover duration among years was assessed via the coefficient of variation (CV) among 10-yr moving windows for each lake and among the GLV.

For diagnostics, we examined residuals of resulting models for normality and equality of variance (Fox 2008). For the GLV-specific models, we checked for coherence in ice phenology among GLV lakes by determining the correlation in ice duration by year. For all models, we tested for remaining temporal autocorrelation in the residuals using the "acf" function in R (R Core Team 2019) and verified a lack of collinearity among predictors by calculating variance inflation factors, for which we considered values below 5 to indicate suitability (Akinwande et al. 2015).

## Results

#### **Green Lakes Valley**

Since 1983, the ice-free period in the GLV has increased by an average of 24 d (LME,  $\beta = 0.66$ , SE = 0.12, p < 0.001; Table 1; Fig. 2), with a range of 17–32 d (Table 1). This pattern was driven by both a 9-d average advance in ice-off date (LME,  $\beta = -0.25$ , SE = 0.09, p = 0.006) and especially a 15-d average delay in ice-on date (LME,  $\beta = 0.41$ , SE = 0.07, p < 0.001; Table 1; Fig. 2). Lake elevation negatively affected open water duration (LME,  $\beta = -0.23$ , SE = 0.03, p < 0.001), such that the highest lakes (e.g., GL4 and GL5) were typically ice-free for 41 fewer days than the lower lakes (GL1-3). Although there was no significant interaction between elevation and year (LME,  $\beta = 0.002$ , SE = 0.001, p = 0.20), estimates of the temporal trend for lakes individually highlight variation in magnitude (Table 1). Model coefficients of determination  $(R^2)$  were 0.62 (marginal) and 0.65 (conditional upon the random effects). Lastly, from the linear analysis, ice thickness in GL4 decreased at a rate of  $0.88 \text{ cm yr}^{-1}$  (Fig. 2), declining from an average 127 cm during the first three observations on record to an average 94 cm by the last three observations (26% decrease). However, the break-point analysis detected a break point occurring in 2000, with little change in ice thickness from 2000 to 2019 (Fig. 2).

Over the 36-yr time series, the average ice-cover duration of lakes in the GLV was 246  $\pm$ 26 d, which ranged from 273 d in GL5 to 226 d in GL1. Among years, ice-cover duration for lakes in the GLV were moderately coherent (r = 0.78), with particularly high correlations for GL1–3 (r = 0.89–0.94) and between GL4 and GL5 (r = 0.96). Across all lakes, the interannual heterogeneity (CV) in ice-cover duration decreased slightly (~ 4%) over this time period.

#### Northern Hemisphere comparisons

Over the same time period (1983–2013), our analysis of lakes in the Northern Hemisphere revealed a 13-d average decrease in ice-cover duration (LME,  $\beta = -0.41$ , SE = 0.05, p < 0.001; Table 1), with positive effects of elevation (LME,  $\beta = 0.09$ , SE = 0.008, p < 0.001) and latitude (LME,  $\beta = 3.72$ , SE = 0.20, p < 0.001), reflecting the important influence of

**Table 1.** Coefficients (standard error) for all linear mixed effects models of lakes in the Northern Hemisphere and Green Lakes Valley (GLV; lakes 1–5: GL1–5) datasets. The full GLV dataset is analyzed (1983–2018) for lake ice-free duration. A truncated dataset (1983–2013) is also used for comparing ice-cover duration in the GLV to the Northern Hemisphere dataset and includes lake elevation, latitude, and respective interactions, while also accounting for random slopes (year) within the model. All parameters are significant to p < 0.05 unless noted as "NS."

|  |                       | Year        | Elevation<br>(m ASL) | Year ×<br>elevation | Latitude<br>(°N) | Year<br>× latitude | Elevation<br>× latitude |
|--|-----------------------|-------------|----------------------|---------------------|------------------|--------------------|-------------------------|
| Green Lakes<br>Valley 1983–2018                | lce-on                | 0.41(0.07)  |                      |                     |                  |                    |                         |
|  | Ice-off               | -0.25(0.09) |                      |                     |                  |                    |                         |
|  | Ice-free duration     | 0.66(0.12)  |                      |                     |                  |                    |                         |
|  |                       | 0.66(0.12)  | -0.23(0.03)          |                     |                  |                    |                         |
|  |                       | 0.66(0.12)  | -0.24(0.03)          | NS                  |                  |                    |                         |
|  | GL1 ice-free duration | NS          |                      |                     |                  |                    |                         |
|  | GL2 ice-free duration | 0.55(0.27)  |                      |                     |                  |                    |                         |
|  | GL3 ice-free duration | 0.54(0.25)  |                      |                     |                  |                    |                         |
|  | GL4 ice-free duration | 0.86(0.26)  |                      |                     |                  |                    |                         |
|  | GL5 ice-free duration | 0.84(0.26)  |                      |                     |                  |                    |                         |
| Comparisons<br>1983–2013<br>ice-cover duration | Northern Hemisphere   | -0.45(0.05) |                      |                     |                  |                    |                         |
|  |                       | -0.46(0.05) | 0.03(0.01)           |                     |                  |                    |                         |
|  |                       | -0.45(0.05) | 0.03(0.01)           | NS                  |                  |                    |                         |
|  |                       | -0.43(0.05) |                      |                     | 2.61(0.23)       |                    |                         |
|  |                       | -0.42(0.05) |                      |                     | 2.76(0.24)       | -0.02(0.005)       |                         |
|  |                       | -0.41(0.05) | 0.09(0.008)          |                     | 3.71(0.20)       |                    |                         |
|  |                       | -0.41(0.05) | 0.09(0.008)          |                     | 3.72(0.20)       |                    | 0.003(0.001)            |
|  | Green Lakes Valley    | -0.66(0.16) |                      |                     |                  |                    |                         |
|  | 2                     | -0.66(0.16) | 0.24(0.03)           |                     |                  |                    |                         |
|  |                       | -0.66(0.16) | 0.24(0.03)           | NS                  |                  |                    |                         |



**Fig 2.** Ice clearance (**A**) and ice formation (**B**) ordinal dates, as well as ice-cover duration (**C**) among Green Lakes 1–5 (elevation: 3425-3620 m ASL) in the Green Lakes Valley, Colorado from 1983 to 2018. Annual maximum ice thickness in Green Lake 4 from 1982 to 2018 is also included (**D**). A linear trend is shown for all plots except ice thickness, for which was fit with a break-point analysis demonstrating a break point in 2000.



**Fig 3.** Estimates from a random slope model representing the number of days per year decrease in ice-cover duration for each lake in the Northern Hemisphere dataset as well as for the Green Lakes Valley, Colorado; GLV from 1983 to 2013, with each lake's latitude and elevation shown for comparison. Note that lakes in the GLV are the highest elevation but nearly the lowest latitude lakes, while generally, the highest latitude lakes of the Northern Hemisphere dataset are among the lowest elevation.

cold climates. There were also significant interactions between year and latitude (LME,  $\beta = -0.02$ , SE = 0.005, p < 0.001) and between elevation and latitude (LME,  $\beta = 0.003$ , SE = 0.001,

p < 0.001), but not between year and elevation. However, the highest latitude lakes in the Northern Hemisphere dataset were also generally the lowest elevation (Fig. 3). Model

coefficients of determination were 0.54 (marginal) and 0.84 (conditional upon the random effects).

By comparing the patterns for GLV with those of Northern Hemisphere lakes generally, we found that the decreasing trend in ice-cover duration for GLV (20 d) was ~ 50% faster than the Northern Hemisphere average (13 d), with ice-cover duration for lakes in the GLV decreasing faster than 87% of all lakes in the Northern Hemisphere dataset. On average, the GLV is ice-free for 121 d per year compared to an average of 211 d across the Northern Hemisphere. Stated another way, this means the GLV experienced a 20% increase in ice-free days between 1983 and 2013 compared to an 8% increase for the Northern Hemisphere dataset.

## Discussion

Ice phenology in the GLV of the Southern Rocky Mountains has shifted considerably over the last 36 yr. On average, the total annual ice-free season has increased by approximately 3 weeks since 1983. Because these small, highelevation systems are frozen for > 225 d per year, this change amounts to a 20% increase in the open-water period. Statistically, this pattern was driven more strongly by shifts in the timing of ice-on rather than ice-off, with ice formation being delayed at nearly twice the rate of earlier ice clearance. By applying the same analytical approach to well-studied lakes in the Northern Hemisphere sampled over a matching time frame, we found that while ice-cover duration is decreasing for lakes broadly, consistent with past analyses (Brown and Duguay 2010), the annual rate of ice cover loss in the highelevation lakes of the GLV (0.66 d yr<sup>-1</sup>) is occurring ~ 50% faster than the Northern Hemisphere average  $(0.41 \text{ d yr}^{-1})$ . Although increasing latitude is associated with large decreases in ice-cover duration (Latifovic and Pouliot 2007), our analyses highlight how rates of changing ice cover concurrently vary with elevation. Due to the short growing season inherent to mountain lakes, the proportional increase in ice-free duration is 2.5 times larger in the GLV compared to the Northern Hemisphere average.

The current results are broadly consistent with previous analyses of alpine lake ecosystems. Our estimated trend in ice clearance ( $-0.25 \text{ d yr}^{-1}$ ) is similar to another study of the GLV over a shorter time period ( $-0.21 \text{ d yr}^{-1}$ ; Preston et al. 2016). Comparing across the Northern Hemisphere, Lake Lunz (608 m ASL) in Austria demonstrated a decrease in ice-cover duration of 0.35 d yr<sup>-1</sup> from 1921 to 2015 (Kainz et al. 2017), whereas a lake in the Tatra Mountains, Poland (1935 m ASL) decreased ice cover at a rate of 0.7 d yr<sup>-1</sup> from 1971 to 2007 (Choinski et al. 2010). Further, a study on the Tibetan plateau (mean: 4500 m ASL) found that ice cover decreased in some lakes from 2001 to 2010 by an average 0.65 d yr<sup>-1</sup> (Kropacek et al. 2013). With respect to high-latitude lakes, Latifovic and Pouliot (2007) reported that boreal lakes in the Canadian Arctic exhibited both an earlier break-up (0.99 d yr<sup>-1</sup>) and later

Mountain lake ice phenology

freeze-up (0.76 d yr<sup>-1</sup>) between 1985 and 2004 (Latifovic and Pouliot 2007). Overall, however, studies of changes in ice-cover duration for high-elevation systems—which can be variably defined among regions—remain relatively rare, and comparisons among studies can be highly sensitive to the sampling period and duration (Kainz et al. 2017), underscoring the value of the standardized comparisons among Northern Hemisphere lakes conducted here.

Mountain lakes are dynamic systems that exhibit and integrate numerous facets of a changing environment. The decreasing trend in maximum ice thickness in GL4 may be slowing given that Caine (2002) reported maximum ice thickness decreasing at a rate of 2 cm  $yr^{-1}$  (SE: 0.21) compared to our estimate of 0.88 cm  $yr^{-1}$  (SE: 0.17) in GL4. It is possible that permafrost melt, which was detected by a state shift of SO<sub>4</sub><sup>-</sup> in 2000, may be causing the detected break point and subsequent weakening in decreasing ice thickness (Caine 2010). It is also possible another state shift could occur once this permafrost buffer expires. Other factors may also be acting upon lakes to produce varying results. Others have shown how lake morphometry, cloud cover, geography, and catchment characteristics (Brown and Duguay 2010), can influence ice phenology trends over large scales; such changes can also be nonlinear (Supplemental Material). In the GLV, it is possible that differing lake volumes, flushing rates, and topographic shading may be influencing differing rates of change. Comparatively, the lakes in the GLV are small and shallow compared to those of the Northern Hemisphere dataset, potentially amplifying ice-loss effects in the GLV. Also, decreased water clarity derived from chlorophyll (Chl) a or dissolved organic matter (DOM) has been linked to increases in lake temperature (Pilla et al. 2018), and this could have implications for ice phenology. In the GLV specifically, Preston et al. (2016) further showed that Chl a concentrations correlate positively with earlier ice out. It is also possible that elevational advancement of terrestrial vegetation (specifically, Salix glauca), which has been detected in the GLV based on aerial imagery between 1972 and 2008 (Bueno de Mesquita et al. 2018), could enhance inputs of DOM into the aquatic ecosystem, leading to decreased clarity. Finally, for lakes that are distributed within highly populated areas, which is common, we highlight the potential for human activity to further alter lake water quality through shifts in nutrients, ion concentrations, and dissolved organic carbon (Howell et al. 2012).

Our findings suggest that, relative to other lakes with ice phenology data in the Northern Hemisphere, lake ice-loss is occurring at faster rates in alpine lakes in the Rocky Mountains. Such changes in ice cover have potential implications for within-lake processes and both the quality and quantity of downstream water resources. Mountain lakes are sensitive systems that provide water for half of the world's population (Woodwell 2004). As water resource demands increase under a growing global population (Vorosmarty et al. 2000), the quality and quantity of available water will be at the forefront of societal importance. We emphasize the need to further examine understudied mountain lakes and their role as a water resource globally.

## References

- Adrian, R., N. Walz, T. Hintze, S. Hoeg, and R. Rusche. 1999. Effects of ice duration on plankton succession during spring in a shallow polymictic lake. Freshw. Biol. **41**: 621–632. doi:10.1046/j.1365-2427.1999.00411.x.
- Akinwande, M. O., H. G. Dikko, and A. Samson. 2015. Variance inflation factor: As a condition for the inclusion of suppressor variable(s) in regression analysis. Open J. Stat. 5: 754–767. doi:10.4236/ojs.2015.57075.
- Arvola, L., and others. 2010. The impact of the changing climate on the thermal characteristics of lakes, p. 85–101. *In* G. George [ed.], *The impact of climate change on European lakes*. Aquatic Ecology Series. The Netherlands: Springer. doi:10.1007/978-90-481-2945-4\_6.
- Bahls, P. 1992. The status of fish populations and management of high mountain lakes in the western United States. Northwest Sci. **66**: 3.
- Bates, D., M. Maechler, B. Bolker, and S. Walker. 2015. Fitting linear mixed-effects models using lme4. J. Stat. Softw. 67: 1–48. doi:10.18637/jss.v067.i01.
- Benson, B., J. Magnuson, and S. Sharma. 2000, *Global Lake and River Ice Phenology Database, Version 1*. Boulder, Colorado USA. NSIDC: National Snow and Ice Data Center. doi: 10.7265/NSW66HP8.
- Brown, L. C., and C. R. Duguay. 2010. The response and role of ice cover in lake-climate interactions. Prog. Phys. Geogr. 34: 671–704. doi:10.1177/0309133310375653.
- Bueno de Mesquita, C. P., L. S. Tillmann, C. D. Bernard, K. C. Rosemond, N. P. Molotch, and K. N. Suding. 2018. Topographic heterogeneity explains patterns of vegetation response to climate change (1972-2008) across a mountain landscape, Niwot Ridge, Colorado. Arct. Antarct. Alp. Res. 50: e1504492. doi:10.1080/15230430.2018.1504492.
- Caine, N. 2001. Geomorphic systems of Green Lakes Valley, p. 45–74. *In Alpine dynamics: The structure and function of an alpine ecosystem: Niwot Ridge, Colorado*. Oxford Univ. Press.
- Caine, N. 2002. Declining ice thickness on an alpine lake is generated by increased winter precipitation. Clim. Change **54**: 463–470. doi:10.1023/A:1016111810814.
- Caine, N. 2010. Recent hydrologic change in a Colorado alpine basin: An indicator of permafrost thaw? Ann. Glaciol. **51**: 130–134. doi:10.3189/172756411795932074.
- Caine, T. 2018. Lake ice clearance and formation data for Green Lakes Valley from 1968 ongoing. Environmental Data Initiative. doi:10.6073/pasta/84e39dccd8439f0f3e320 f60f4c42759.
- Caldwell, T. J., S. Chandra, K. Feher, J. B. Simmons, and Z. Hogan. 2020. Ecosystem response to earlier ice break-up

date: Climate-driven changes to water temperature, lake-habitat-specific production, and trout habitat and resource use. Glob. Chang. Biol. **26**: 5475–5491. doi:10.1111/gcb. 15258.

- Catalan, J., and J. C. Donato-Rondón. 2016. Perspectives for an integrated understanding of tropical and temperate high-mountain lakes. J. Limnolol. **75**: 215–234. doi:10. 4081/jlimnol.2016.1372.
- Christianson, K. R., B. M. Johnson, and M. B. Hooten. 2020. Compound effects of water clarity, inflow, wind and climate warming on mountain lake thermal regimes. Aquat. Sci. **82**: 1–17. doi:10.1007/s00027-019-0676-6.
- Christianson, K. R., and B. M. Johnson. 2020. Combined effects of early snowmelt and climate warming on mountain lake temperatures and fish energetics. Arct. Antarct. Alp. Res. **52**: 130–145. doi:10.1080/15230430.2020. 1741199.
- Choinski, A., L. Kolendowicz, J. Pociask-Karteczka, and L. Sobkowiak. 2010. Changes in lake ice cover on the Morskie Oko Lake in Poland (1971 2007). Adv. Clim. Chang. Res. **1**: 71–75. doi:10.3724/SP.J.1248.2010.00071.
- Fox, J. 2008, *Applied regression analysis and generalized linear* models, 2nd Edition. SAGE.
- Henneman, H. E., and H. G. Stefan. 1999. Albedo models for snow and ice on a freshwater lake. Cold Reg. Sci. Technol. 29: 31–48. doi:10.1016/S0165-232X(99)00002-6.
- Howell, E. T., K. M. Chomicki, and G. Kaltenecker. 2012. Patterns in water quality on Canadian shores of Lake Ontario: Correspondence with proximity to land and level of urbanization. J Great Lakes Res. **38**: 32–46. doi:10.1016/j.jglr. 2011.12.005.
- Jacobson, P. C., T. S. Jones, P. Rivers, and D. L. Pereira. 2008. Field estimation of a lethal oxythermal niche boundary for adult ciscoes in Minnesota lakes. Trans. Am. Fish. Soc. 137: 1464–1474. doi:10.1577/T07-148.1.
- Kainz, M. J., R. Ptacnik, S. Rasconi, and H. H. Hager. 2017. Irregular changes in lake surface water temperature and ice cover in subalpine Lake Lunz, Austria. Inland Waters 7: 27–33. doi:10.1080/20442041.2017.1294332.
- Kropacek, J., F. Maussion, F. Chen, S. Hoerz, and V. Hochschild. 2013. Analysis of ice phenology of lakes on the Tibetan Plateau from MODIS data. Cryosphere 7: 287–301. doi:10.5194/tc-7-287-2013.
- Latifovic, R., and D. Pouliot. 2007. Analysis of climate change impacts on lake ice phenology in Canada using the historical satellite data record. Remote Sens. Environ. **106**: 492–507. doi:10.1016/j.rse.2006.09.015.
- Lopez, L. S., B. A. Hewitt, and S. Sharma. 2019. Reaching a breaking point: How is climate change influencing the timing of ice breakup in lakes across the northern hemisphere? Limnol. Oceanogr. 64: 2621–2631. doi:10.1002/lno.11239.
- Muggeo, V. M. R. 2020. segmented. R package version 1.2-0.
- Pilla, R. M., C. E. Williamson, J. Zhang, R. L. Smyth, J. D. Lenters, J. A. Bentrup, L. B. Knoll, and T. J. Fisher. 2018.

Browning-related decreases in water transparency lead to long-term increases in surface water temperature and thermal stratification in two small lakes. J. Geophys. Res. Biogeo. **123**: 1651–1665. doi:10.1029/2017JG004321.

- Preston, D. L., N. Caine, D. M. McKnight, M. W. Williams, K. Hell, M. P. Miller, S. J. Hart, and P. T. J. Johnson. 2016. Climate regulates alpine lake ice cover phenology and aquatic ecosystem structure: Climate and alpine lakes. Geophys. Res. Lett. 43: 5353–5360. doi:10.1002/2016GL069036.
- R Core Team. 2019. *R: A language and environment for statistical computing*. R Foundation for Statistical Computing, Vienna, Austria. http://www.R-project.org/
- Sharma, S., and others. 2019. Widespread loss of lake ice around the Northern Hemisphere in a warming world. Nat. Clim. Change Lett. 9: 227–231. doi:10.1038/s41558-018-0393-5.
- Vorosmarty, C. J., P. Green, J. Salisbury, and R. B. Lammers. 2000. Global water resources: Vulnerability from climate change and population growth. Science 289: 284–288. doi: 10.1126/science.289.5477.284.
- Williams, S. G., and H. G. Stefan. 2006. Modeling of lake ice characteristics in North America using climate, geography,

and lake bathymetry. J. Cold Reg. Eng. **20**: 140–167. doi: 10.1061/(ASCE)0887-381X(2006)20:4(140).

- Williamson, C. E., J. E. Saros, W. F. Vincent, and J. P. Smol. 2009. Lakes and reservoirs as sentinels, integrators, and regulators of climate change. Limnol. Oceanogr. 54: 2273–2282. doi:10.4319/lo.2009.54.6\_part\_2.2273.
- Woodwell, G.M. 2004. Mountains: Top down. Ambio, Special Report 13. The royal colloquium: Mountain areas: A global resource. 35–38.

#### Acknowledgments

Data from this study were collected through the Niwot Ridge Long Term Ecological Research program (https://nwt.lternet.edu/). We thank Jen Morse, Henry Brandes, and Dillon Ragar for data collection and the CU Mountain Research Station. Funding was provided by the University of Colorado, Niwot Ridge LTER program (NSF DEB-1637686). The authors do not declare any conflicts of interest.

Submitted 02 March 2020 Revised 09 December 2020 Accepted 21 December 2020